

# **JGR** Atmospheres

# **RESEARCH ARTICLE**

10.1029/2021JD036355

#### **Key Points:**

- Ground-based observations show variations in cloud droplet number (+100%) and their radius (-14%) associated with high phytoplankton activity
- The biota-induced variations in stratiform cloud microphysics are comparable in magnitude to those generated by anthropogenic inputs
- Cloud optical properties mainly respond to meteorology, however, the increase in summer droplet number contributes to enhanced albedo

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

K. Mansour and M. Rinaldi, k.mansour@isac.cnr.it; m.rinaldi@isac.cnr.it

#### Citation:

Mansour, K., Rinaldi, M., Preißler, J., Decesari, S., Ovadnevaite, J., Ceburnis, D., et al. (2022). Phytoplankton impact on marine cloud microphysical properties over the Northeast Atlantic Ocean. *Journal of Geophysical Research: Atmospheres*, *127*, e2021JD036355. https://doi.org/10.1029/2021JD036355

Received 13 DEC 2021 Accepted 30 APR 2022

#### **Author Contributions:**

Conceptualization: Karam Mansour, Matteo Rinaldi Investigation: Karam Mansour, Matteo Rinaldi, Jana Preißler, Jurgita Ovadnevaite, Darius Ceburnis Methodology: Karam Mansour, Matteo Rinaldi Supervision: Matteo Rinaldi, Stefano Decesari, Maria C. Facchini, Colin O'Dowd Visualization: Karam Mansour

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Phytoplankton Impact on Marine Cloud Microphysical Properties Over the Northeast Atlantic Ocean

Karam Mansour<sup>1,2</sup>, Matteo Rinaldi<sup>1</sup>, Jana Preißler<sup>3</sup>, Stefano Decesari<sup>1</sup>, Jurgita Ovadnevaite<sup>4</sup>, Darius Ceburnis<sup>4</sup>, Marco Paglione<sup>1</sup>, Maria C. Facchini<sup>1</sup>, and Colin O'Dowd<sup>4</sup>

<sup>1</sup>Italian National Research Council - Institute of Atmospheric Sciences and Climate (CNR-ISAC), Bologna, Italy, <sup>2</sup>Oceanography Department, Faculty of Science, Alexandria University, Alexandria, Egypt, <sup>3</sup>Leosphere, Saclay, France, <sup>4</sup>School of Physics, Ryan Institute's Centre for Climate and Air Pollution Studies, National University of Ireland Galway, Galway, Ireland

**Abstract** The current understanding of the impact of natural cloud condensation nuclei (CCN) variability on cloud properties in marine air is low, thus contributing to climate prediction uncertainty. By analyzing cloud remote sensing observations (2009–2015) at Mace Head (west coast of Ireland), we show the oceanic biota impact on the microphysical properties of stratiform clouds over the Northeast Atlantic Ocean. During spring to summer (seasons of enhanced oceanic biological activity), clouds typically host a higher number of smaller droplets resulting from increased aerosol number concentration in the CCN relevant-size range. The induced increase in cloud droplet number concentration (+100%) and decrease in their radius (-14%) are comparable in magnitude to that generated by the advection of anthropogenically influenced air masses over the background marine boundary layer. Cloud water content and albedo respond to marine CCN perturbations with positive adjustments, making clouds brighter as the number of droplets increases. Cloud susceptibility to marine aerosols overlaps with a large variability of cloud macrophysical and optical properties primarily affected by the meteorological conditions. The above findings suggest the existence of a potential feedback mechanism between marine biota and the marine cloud-climate system.

**Plain Language Summary** Marine phytoplankton contributes to the natural aerosol burden in the atmosphere. Aerosol particles serve as cloud condensation nuclei that play an important role in cloud formation. Clouds cool the earth's surface by scattering and reflecting the incoming solar radiation to space. We found that the enhancement of oceanic biological activity during spring and summer times leads to an increase in the number of droplets in stratiform clouds, over the Northeast Atlantic Ocean, and a decrease in their size. This happens as a consequence of the higher aerosol concentrations resulting from enhanced marine biogenic emissions. The magnitude of changes in cloud microphysical parameters caused by contrasting marine biota conditions is comparable to that observed by comparing anthropogenically influenced and background marine air masses. Furthermore, we observed that cloud brightness responds positively to marine aerosol sources (*i.e.*, clouds are brighter as the number of droplets increases), even though this relationship is non-linear. These results are important for understanding the causes of variability in cloud radiative effects and for the improvement of climate predictions.

# 1. Introduction

Stratiform clouds mostly originate over the oceans and have a profound impact on the Earth's radiation budget. Therefore, understanding the causes of the variability in their radiative effects is necessary for a better comprehension of the Earth's energy budget and climate forecast. The future effort to reduce the impact of greenhouse gases and their associated global warming potential puts the emphasis on natural sources (Boucher et al., 2013; Satheesh & Moorthy, 2005) and processes controlling marine stratiform clouds.

The radiative impact and brightness of clouds strongly depend on their microphysical properties (Falkowski et al., 1992) such as liquid water content (LWC), cloud droplet effective radius ( $R_{eff}$ ), and cloud droplet number concentration (CDNC). At constant LWC, the increase of aerosol number and cloud condensation nuclei (CCN) atmospheric concentrations results in higher CDNC while the  $R_{eff}$  diminishes. Therefore, clouds are optically brighter as more solar radiation is scattered upward and less is transmitted through. This is known as the first



Writing – original draft: Karam Mansour, Matteo Rinaldi, Stefano Decesari

Writing – review & editing: Karam Mansour, Matteo Rinaldi, Stefano Decesari, Jurgita Ovadnevaite, Darius Ceburnis, Marco Paglione, Maria C. Facchini, Colin O'Dowd aerosol-cloud indirect effect (Twomey effect; Twomey [1974]) and represents the main factor regulating the instantaneous radiative forcing from aerosol-cloud interaction (RFaci). The induced changes of cloud microphysics can affect the onset of precipitation, cloud water path, and cloud cover according to a range of adjustment mechanisms which may either amplify or dampen the ACI radiative effect (Albrecht, 1989; Rosenfeld et al., 2019) and that may be as important as the instantaneous effects. Such mechanisms have been a subject of intensive research in the case of marine stratus and stratocumulus clouds which represent one of the most common cloud types (in terms of relative cloud fraction) over the Earth. In particular, the oceanic regions covered by stratocumulus clouds exhibit a higher sensitivity of cloud albedo to aerosol perturbations (Gryspeerdt et al., 2019). Aerosol-cloud interactions are challenging to investigate because the cloud response to changes in aerosol concentration and composition is intertwined with the response to variations in the meteorological conditions (Loeb & Schuster, 2008). Therefore, it is difficult to separate aerosol effects from meteorological impacts (Sato & Suzuki, 2019; Stevens & Feingold, 2009). In the marine boundary layer (MBL), cloud cover is generally associated with lower troposphere stability (LTS; Janssen et al. [2011]; Wood & Bretherton [2006]), a measure of convection strength. A larger cloud fraction is generally associated with higher LTS (Andersen et al., 2017; Klein & Hartmann, 1993). The transport and mixing of aerosols from the surface to the cloud base are essential since they influence the proportion of particles that activate into cloud droplets (Janssen et al., 2011). Excluding the meteorological impact, Rosenfeld et al. (2019) concluded that aerosol-driven droplet concentrations account for 75% of the radiative cooling effect variability, through their effects on the coverage of low-level ocean clouds.

Marine biota is a major source of biogenic aerosols in the atmosphere, through both primary and secondary aerosol formation pathways. The present limited knowledge of such natural emissions contributes to the high uncertainties in aerosol-cloud forcing estimates (Carslaw et al., 2013; Rap et al., 2013). Most of the research done so far in this regard is focused on the Southern Ocean, which is considered a unique environment because of its remoteness. In such conditions, natural climate interactions can be more easily investigated, without interference from anthropogenic sources (McCoy et al., 2015). Cloud droplets have been often reported to be more numerous and smaller in size over oceanic regions characterized by high biological activity (HBA) than over low biological activity (LBA) regions (McCoy et al., 2015; Meskhidze & Nenes, 2006; Sorooshian & Duong, 2010; Sorooshian et al., 2009). Laboratory experiments and field observations have elucidated that the increase in CCN concentration is linked to phytoplankton abundance (Collins et al., 2016; Fossum et al., 2018, 2020; Mayer et al., 2020; Sellegri et al., 2021), where the effect of human activities appears to be minimal (Falkowski et al., 1992). Contrasting conclusions were presented by Miller and Yuter (2008) who did not always find a correlation between oceanic biological activity and Reff, questioning the existence of a causal link between secondary aerosols and marine shallow cloud properties. This ambiguity may reflect the complexity of studying marine biogenic aerosol-cloud interactions, while at the same time some studies may have been biased by dominant continental aerosols as noted by Meskhidze and Nenes (2010).

In this work, we study the CCN variability and ocean biota impacts on cloud properties in unperturbed conditions over the Northeast Atlantic (NEA) Ocean. The NEA Ocean exhibits a variety of seasonal phytoplankton biomass patterns, with a distinct bloom beginning mostly in spring and continuing through the summer (Friedland et al., 2016; Lacour et al., 2015). Such biological activity has been proven to impact the chemical composition, number concentration, and size distribution of marine aerosols (Ault et al., 2013; Facchini et al., 2008; O'Dowd et al., 2004; Prather et al., 2013), which would eventually change cloud characteristics (Charlson et al., 1987). In this region, a few studies have reported a link between oceanic biological activity and cloud properties. For instance, an increase of about 15%–100% in CDNC was attributed to the impact of marine organic aerosols by using a mono-dimensional cloud droplet model (O'Dowd et al., 2004). Sea spray enriched with primary organics enhances the CCN activation efficiency leading to high CDNC concentration during less than 6 hr of cloud observations (Ovadnevaite et al., 2011). Such influence was evident in the summer when biological activity is maximized (Gantt et al., 2012). On the contrary, some studies lessened the relative importance of the sea-spray organic fraction in determining its CCN properties (Hendrickson et al., 2021), suggesting that non-sea-salt sulfate may be the strongest link between marine biology and CCN (Quinn et al., 2019; Saliba et al., 2020; Sanchez et al., 2018).

In addition to the complexity of cloud properties adjustment and feedback processes affecting the sign and magnitude of ACI radiative effects, there is also a methodological problem in the assessment of such effects from the observational point of view (Gryspeerdt et al., 2019), as most common remote sensing techniques cannot provide full vertical profiling of all relevant properties for both clouds and aerosols. Cloud observations are 21698996, 2022, 10, Downloaded from https://agupubs.



usually carried out at the cloud top, with the microphysics of the main entrainment zone of aerosol-rich air (at cloud base) being inaccessible. In this study, we employ a combination of ground-based aerosol in situ and cloud remote sensing observations to overcome the retrieval limitations of satellite-based techniques while providing much wider temporal coverage (ca. 6 years of observations) with respect to in situ measurements available from aircraft missions (Sinclair et al., 2020).

The Mace Head (MHD) global atmosphere watch research station  $(53.33^{\circ}N, 09.90^{\circ}W; 21 \text{ m a.s.l})$ , located about 80 m from the waterline, on the west coast of Ireland, is exposed to the North Atlantic Ocean through the westerly sector. Stratiform clouds at MHD have a prevailing marine origin. Numerous studies of in situ observations of clouds have been done through the MHD remote sensing unit, including the development of the SYnergistic Remote Sensing Of Clouds (SYRSOC) algorithm (Martucci & O'Dowd, 2011). This retrieval algorithm was further extensively used to combine in situ ground aerosol and cloud measurements (Martucci et al., 2013; Ovadnevaite et al., 2011). The marine and continental stratiform clouds were characterized by Preißler et al. (2016) who found that a lower CDNC with broader  $R_{eff}$  during clean marine conditions than in continental cases, in line with the comprehensive in situ database of stratus cloud (Miles et al., 2000).

In this study, we assess the relationship between oceanic biological activity and cloud properties in background marine conditions, over the NEA Ocean, by exploring a unique dataset of multi-year cloud observations (from February 2009 to January 2015) measured by ground-based remote sensing instruments at MHD and retrieved by SYRSOC method. The data include cloud microphysical (CDNC,  $R_{eff}$ , LWC), cloud macrophysical (cloud base height [ $H_{base}$ ], cloud top height [ $H_{top}$ ], and cloud thickness [ $H_{thick}$ ]), and cloud optical properties (albedo and cloud optical depth [COD]). Cloud observations were analyzed together with satellite ocean color data (sea surface chlorophyll-a concentrations (CHL) as a proxy of phytoplankton activity), air mass back-trajectories and in situ aerosol particle number concentration ( $N_a$ ) data in the CCN relevant size range between 40 and 300 nm in diameter (Quinn & Bates, 2011). Finally, using ECMWF-ERA5 reanalysis atmospheric data, we analyze the main meteorological drivers for stratiform cloud formation in this region. Based on that, we discriminate the effect of meteorology on cloud properties from that of ACI processes.

## 2. Materials and Methods

#### 2.1. Ground-Based Remote Sensing Measurements

The ground-based remote sensing (GBRS) division at MHD, on the west coast of Ireland, has been a Cloudnet station since 2009 (Illingworth et al., 2007) and comprises cloud radar, ceilometer, and microwave radiometer (MWR), and since 2015 a Doppler wind lidar. GBRS provides continuous monitoring of the atmosphere at one location, with high vertical and temporal resolutions. This provides useful detailed insights into cloud processes, which can be a powerful tool for the detection and quantification of cloud indirect effects (Feingold et al., 2003).

The radar is a MIRA36, a 35.5 GHz Ka-band Doppler cloud radar (Melchionna et al., 2008) that measures in-cloud reflectivity, and linear depolarization ratio, and vertical speed of cloud droplets. The radar was also used to track the top level of clouds. Further description of the calibration offset of the radar reflectivity can be found in Preißler et al. (2016).

The ceilometer is a CHM15k measuring 1,064 nm (Heese et al., 2010; Martucci et al., 2010). It detects photons backscattered from atmospheric targets such as cloud droplets or aerosol particles. Based on the optical depth of the atmosphere, it can detect aerosol particles, as well as clouds up to a certain penetration point. The ceilometer was used to detect the cloud base altitude.

The MWR is a multichannel microwave profiler RPG-HATPRO (Crewell & Lohnert, 2003; Lohnert & Crewell, 2003; Lohnert et al., 2009) measuring near water vapor and oxygen absorption lines (Martucci & O'Dowd, 2011).

The SYRSOC retrieval method uses GBRS data from the three abovementioned instruments as a primary input. It allows for vertically resolved determination of cloud properties from the ground. Using the SYRSOC algorithm (Martucci & O'Dowd, 2011; Preißler et al., 2016), the microphysical, macrophysical, and optical properties are retrieved. Homogeneous sections of less than 1-hr duration have been selected of all non-precipitating single-layer water clouds. The microphysical properties of each cloud case are represented by a 2D matrix with time as the *x*-axis and height as the *y*-axis. The vertical and temporal resolutions are 15 m and 10 s.



Starting from the 6 years (February 2009 to January 2015) of SYRSOC dataset cloud observations (Prei $\beta$ ler, 2016), the air mass back-trajectories were calculated using the Hybrid Single-Particle Lagrangian Integrated Trajectory (HYSPLIT4) model, a transport and dispersion model developed by the National Oceanic and Atmospheric Administration (NOAA), Air Resources Laboratory (ARL; Rolph et al. [2017]; Stein et al. [2015]) to define the marine source areas, coming from the Atlantic Ocean. Based on back-trajectories (Figure S1 in Supporting Information S1), a total of 52 clean marine stratiform clouds were classified as a representative of the North Atlantic Ocean background and used in this study. The cloud cases are distributed over 47 days. The ground level average equivalent black carbon measured in situ by a multi-angle absorption photometer (O'Dowd et al., 2014) associated with the considered 52 cases is  $8.8 \pm 5.2$  ng/m<sup>3</sup>, proving negligible anthropogenic contribution.

The cloud microphysics retrievals from SYRSOC were compared with in situ aircraft measurements from a 2-day field experiment conducted at MHD (Martucci et al., 2013), showing a very good agreement in the SYRSOC full profiles. Nevertheless, we compared the SYRSOC results of the selected cases with the Moderate-Resolution Imaging Spectroradiometer (MODIS) level-2 cloud product from both Terra (MOD06) and Aqua (MYD06) platforms (Platnick et al., 2017). The comparison is explained in the supplementary materials (see Text S1 in Supporting Information S1) and shown in Figure S2 in Supporting Information S1. The  $R_{eff}$  by SYRSOC is slightly lower than that by MODIS, consistently with the whole dataset (Preißler et al., 2016). CDNC is more scattered, consistently with the fact that it is still a very uncertain parameter (Grosvenor et al., 2018), nevertheless, the data fall overall around the 1:1 line.

#### 2.2. In Situ Measurements

The submicron aerosol size distributions were measured by a Scanning Mobility Particle Sizer ( $0.02 < \text{particle diameter (Dp)} < 0.5 \ \mu\text{m}$ ), where aerosols are neutralized (aerosol neutralizer: Kr-85, TSI Model 3077) and size-discriminated based on their mobility diameter (differential mobility analyzer: TSI Model 3071) and counted by a condensation particle counter (TSI Model 3010). Aerosol sizing instruments were located downstream of Nafion driers (relative humidity <20%; Ovadnevaite et al. [2017]), and thus, size distributions refer to the aerosol dry diameter (Sanchez et al., 2017). We have selected the particle number concentration of 40 < Dp < 300 nm for this study since this diameter range is large enough to take up water vapor and serve as nuclei for cloud droplet formation in MBL (Quinn & Bates, 2011).

#### 2.3. Satellite Ocean Color Data

The present study is based on the best estimates of "Level-4 Cloud Free" satellite ocean-color data products which are the result of merging Sea-viewing Wide Field of View (SeaWiFS), Moderate Resolution Imaging Spectroradiometer (MODIS-Aqua), MEdium Resolution Imaging Spectrometer (MERIS), Visible and Infrared Imager/Radiometer Suite (VIIRS) and Ocean and Land Color Instrument-Sentinel 3A (OLCI-S3A) sensors. The sea surface CHL obtained from the EU Copernicus Marine Environment Monitoring Service (CMEMS) was used as the reference surrogate for tracing the evaluation of marine biological activity (Rinaldi et al., 2013; Yoon et al., 2007) because it is the most widely available and validated ocean color parameter. CHL is commonly used as a proxy of phytoplankton activity in seawater (Behrenfeld et al., 2016; Huot et al., 2007); the more phytoplankton present, the greener the water and the higher CHL. For the NEA Ocean domain (40°–46°N and 46°–0°W), CHL fields were extracted from a global product, available at 1/24° (about 4 km) spatial resolution and daily time-resolution. The analyzed time series covers the period from 2009 to 2015.

#### 2.4. Meteorological Data

The European Centre for Medium-Range Weather Forecasts (ECMWF) fifth generation reanalysis ERA5 (C3S, 2017) data provides estimates for the hourly state of the atmosphere, worldwide, with spatial resolution of  $0.25^{\circ} \times 0.25^{\circ}$  at the surface and different pressure levels. The ERA5 data are available by the ESA Ocean Color-Climate Change Initiative (ESA OC-CCI) team and the Copernicus climate change service (C3S). From the global domain, we extracted the following meteorological parameters in order to evaluate the relationship



between cloud properties and the main meteorological variables in comparison with the importance of the oceanic biological activity.

The air temperature at 2m above sea level (T) and sea surface temperature (SST) was chosen as representative of thermal heating and the relative humidity (RH) was chosen as representative of water vapor abundance in the atmosphere. To represent the atmospheric thermodynamic and dynamic conditions, the lower tropospheric stability (LTS) parameter and the pressure vertical velocity (PVV) were utilized, respectively.

The LTS, an indicator for the air convection, is calculated as the potential temperature difference between 1,000 and 850 hPa (Fuchs et al., 2017; Painemal, 2018), which reflects thermal (in)stability in the sub-cloud layer: the higher the LTS, the more unstable the atmosphere. For the 52 studied marine cloud cases, the LTS ranges from -1.8 to 10.9°C. Positive values of LTS account for 88% of the cases, indicating a dominant thermal instability for the selected cloud cases. The pressure vertical velocity (PVV; Pa s<sup>-1</sup>) is the speed at which the air travels up or down. Negative values of PVV imply upward motion (ascent areas), while positive values represent downward motion (subsidence).

#### 2.5. Spatio-Temporal Correlation Analysis

To investigate the relationship between marine cloud properties and oceanic biological activity (represented by CHL), we implemented the spatio-temporal correlation analysis approach (Mansour, Decesari, Bellacicco, et al., 2020; Mansour, Decesari, Facchini, et al., 2020). The approach is based on evaluating, by standard least squares regression, the correlation coefficients between the GBRS cloud parameters (the average of each variable in the 2D cloud profile; vertically- cloud thickness and horizontally- measurement time), measured at MHD, and sea surface CHL, at each grid point of the NEA domain. The justifications and full details of the approach can be found in the above-cited papers. Basically, we tested the effect of the time-lag on the "cloud properties versus CHL" correlations to compensate for the different phases of the correlated variables. Indeed, surface CHL tracks the growing phase of algal blooms, while the release of phytoplankton exudates, involved in the production of biogenic marine aerosols, occurs mainly at a later stage, during the senescence/demise phase (Miyazaki et al., 2020; Rinaldi et al., 2013). Starting from simultaneous time series, corresponding to "Lag = 0", we shifted CHL backward day by day to calculate the correlation coefficient between the two considered time series. Such a process is conducted for each pixel of the studied domain, to obtain the "Lag = n days" correlation maps (Figures S3-S5 in Supporting Information S1) where "n" is the number of days by which the CHL time-series is shifted back.

#### 2.6. Aerosol-Cloud Interaction Index

To examine the microphysical response of cloud to aerosol loading, the ACI indices for both R<sub>eff</sub> and CDNC were calculated as:

$$ACI_{r} = -\frac{d \ln (R_{eff})}{d \ln (N_{a})}$$
$$ACI_{n} = \frac{1}{3} \frac{d \ln(CDNC)}{d \ln (N_{a})}$$

Both ACI<sub>r</sub> and ACI<sub>n</sub> (Koike et al., 2019; McComiskey & Feingold, 2012; Zhao et al., 2012) emphasize the relative change in the mean layer of  $R_{eff}$  and CDNC to the relative change in the aerosol loading and must be calculated at a constant LWC. The ACI values range from 0 to 0.33 where the lower boundary implies no change in cloud microphysical properties with aerosol variation and the upper boundary suggests a linear relationship.

The ACI must be calculated and compared at constant LWC, due to the dependence of  $R_{eff}$  and CDNC on LWC. To match the time-resolution of aerosol number concentration, the cloud data were averaged vertically from base to top. Then, we used three LWC bins classified as Low, Medium, and High, using the one-third and two-third percentiles as threshold values. The LWC bins are <0.187 (n = 97), 0.187–0.276 (n = 97), and >0.276 gm m<sup>-3</sup> (n = 97).



## 3. Results

#### 3.1. Cloud Properties in Relation to CHL Patterns

Following the spatiotemporal correlation analysis approach (Mansour, Decesari, Bellacicco, et al., 2020; Mansour, Decesari, Facchini, et al., 2020; O'Dowd et al., 2015; Rinaldi et al., 2013), we investigated the spatial distributions of the correlation coefficient between cloud properties observed at MHD and sea surface CHL at each grid point of the NEA domain, considering different time-lags from 0 to 25 days. Among all the analyzed cloud metrics, only  $R_{eff}$  and CDNC demonstrated a robust and significant spatial correlation with CHL, which began to gradually diminish after time-lag of 15 days. The resulting maps are presented in Figures S3 and S4 in Supporting Information S1. The synthetic maps in Figure 1a were created by selecting the maximum correlation obtained at each pixel within the range of time-lags between 0 and 15 days. The spatial distributions of the timelags which maximize the correlation are shown in Figure 1b. In general, a systematic correlation was observed within a specific oceanic region of the NEA domain ( $45^{\circ}$ - $60^{\circ}$ N and  $12^{\circ}$ - $38^{\circ}$ W: black box in Figure 1a), with correlations generally maximized considering a time-lag between 0 and 5 days as indicated in Figure 1b. Overall,  $R_{eff}$  correlated negatively with CHL, while a positive correlation was observed for CDNC.

Importantly,  $N_a$  is positively correlated with CHL in the same identified oceanic region and presents a similar spatio-temporal pattern of the correlation as CDNC and  $R_{eff}$  (Figure 1 and Figure S5 in Supporting Information S1). It follows that the observed correlation between cloud microphysical properties and CHL may indeed evidence a causal link between marine biological activity and cloud microphysical properties. This relation appears to be modulated by the influence that biological activity exerts on the atmospheric concentration of marine aerosol particles. Furthermore, we stress that the high-correlation oceanic region evidenced in this study corresponds closely to the main source of biogenic marine aerosol observed at MHD documented in previous studies (Mansour, Decesari, Facchini, et al., 2020; Rinaldi et al., 2013), further supporting a cause-effect link at the base of the observed correlations.

The correlation coefficient frequency distributions (Figure 1c), taken over the same box region, for time-lags from 0 to 15 days (whereas the frequency distribution up to 25 days is shown in Figure S6 in Supporting Information S1) confirm the correlation between CHL and  $R_{eff}$  (CDNC and  $N_a$ ) is oriented toward negative (positive) values as the time-lag is increased from 0 to 5 days. Furthermore, the percentages of the number of pixels with a negative (positive) significant correlation in that region for  $R_{eff}$  (CDNC and  $N_a$ ) demonstrate that the correlation begins to decline after a time lag of 15 days (Figure S7 in Supporting Information S1). In previous studies, a significant correlation could be observed between cloud-relevant aerosol properties and CHL patterns, with a time-lag between zero and up to 25 days (Mansour, Decesari, Facchini, et al., 2020; O'Dowd et al., 2015). Figure S8 in Supporting Information S1 complements this plot by looking at the correlation as a function of the time lag, considering the average CHL concentration within the high correlation region evidenced in Figure 1a, for all the tested cloud parameters.

Nevertheless, aware that the correlation analysis alone cannot unambiguously demonstrate a cause-effect relationship between phytoplankton patterns and cloud properties, we applied the advanced three-dimensional concentration weighted trajectory (3D-CWT) model (Rinaldi et al., 2021) to the MHD aerosol and cloud data in order to identify the NEA ocean regions associated with higher  $N_a/CDNC$  and lower  $R_{eff}$ . The details of the analysis are presented in the Supporting Information S1 (Text S2). The resulting CWT maps (Figure S9 in Supporting Information S1) show a clear spatial consistency between sources of  $N_a$  and ocean regions associated with high-CDNC/low- $R_{eff}$  clouds, which strengthens the above considerations on the aerosol influence on cloud microphysical properties. Furthermore, these same regions are clearly matching the high correlation region evidenced by the maps shown in Figure 1. To evidence this result, in Figure S10 in Supporting Information S1 we reported how many times a given pixel of the domain presented CWT  $\leq$  median (CWT  $\geq$  median) and negative (positive) significant correlation coefficient, in the case of  $R_{eff}$  (CDNC and  $N_a$ ), by considering the time-lag from 0 to 15 days. This result clearly confirms that oceanic regions characterized by a significant correlation between aerosol and cloud parameters with surface CHL are also dominant sources of marine biogenic aerosol, characterized by the formation of clouds with higher CDNC and lower  $R_{eff}$ .

The reported correlations (CHL vs.  $R_{eff}$ , CDNC, and  $N_a$ ) are driven by the seasonal variability of the considered parameters, as shown in Figure 1d and Table S1 in Supporting Information S1 which evidence that seasons of enhanced sea surface CHL (spring and summer) are also characterized by low  $R_{eff}$  and by high  $N_a$  and CDNC.



# Journal of Geophysical Research: Atmospheres



**Figure 1.** (a) Spatial distributions of the maximum correlation coefficient, within the time-lags from 0 to 15 days, between chlorophyll-a concentrations at each pixel of the Northeast Atlantic ocean and  $R_{eff}$ , cloud droplet number concentration (CDNC) and  $N_a$  measured at Mace Head (MHD). (b) Spatial distribution of the time-lags maximizes the correlation coefficient used in panel (a). Only significant correlation coefficients (p < 0.05) are presented. In (a) and (b) panels, the filled black square corresponds to the MHD station, the black box area comprises grid coordinates  $45^{\circ}$ - $60^{\circ}N$  and  $12^{\circ}$ - $38^{\circ}W$  that indicates a high correlation area, and the spatial resolution of the pixels is roughly 4 km. (c) Correlation coefficient frequency distributions in the identified box at different time-lags from 0 to 15 days (d) Scatter plots between average CHL in the identified black box and average  $R_{eff}$  CDNC, and  $N_a$  for each cloud case; a 3-day time-lag, which maximizes the correlation, was considered. The significant (p < 0.05) correlation coefficients and the best fit line are reported. The vertical bars represent the standard deviation for each cloud case.



Typical LBA conditions, as observed during this study for the NEA Ocean, are characterized by surface CHL in the range of 0.13–0.40 mg m<sup>-3</sup> (min & max over the box in Figure 1a) during the winter season (Dec-Feb). These conditions are typically associated with  $R_{eff}$  and CDNC ranging between 9.6 and 13.2 µm and 16.8–86.3 cm<sup>-3</sup>, respectively. Conversely, the peak of biological productivity was represented by CHL between 0.33 and 0.93 mg m<sup>-3</sup> in summer (Jun-Aug). During this period,  $R_{eff}$  and CDNC ranged 5.4–14.2 µm and 15–213 cm<sup>-3</sup>. In conclusion, CDNC increased from 40 ± 26 to 81 ± 63 cm<sup>-3</sup> [median 28 to 54],  $R_{eff}$  diminished from 12.1 ± 1.2 to 10.3 ± 2.5 µm [median 12.5 to 10.8] while the  $N_a$  ranged from 106 ± 62 cm<sup>-3</sup> to 326 ± 186 [median 78 to 280] cm<sup>-3</sup> going from winter to summer (Table S1 in Supporting Information S1). According to the nonparametric Mann-Whitney U-test for similarity of medians of two independent samples, the differences between winter and summer are statistically significant at p < 0.05 for CHL,  $N_a$ , and  $R_{eff}$  whereas all of the reported variables are significantly different at p < 0.1.

For comparison purposes, Table S2 in Supporting Information S1 also presents a statistical summary of the cloud microphysical parameters obtained by selecting the 24 most polluted cloud cases from the extended dataset (Preißler et al., 2016). These cases represented the advection of air masses from Europe over the NEA Ocean. The mean CDNC of the selected polluted cases was  $209 \pm 146$  cm<sup>-3</sup>, corresponding to an average  $R_{eff}$  of  $8.0 \pm 2.4$  µm. The average equivalent black carbon concentration for the polluted cases was equal to  $231 \pm 219$  ng m<sup>-3</sup>, an order of magnitude higher than the marine background level of less than 15 ng m<sup>-3</sup> (see Discussion for further comparison). It is worth highlighting that the seasonality of polluted cases has an opposite effect of marine cases (i.e., higher CDNC and lower  $R_{eff}$  in winter), as indicated in Figure S11 in Supporting Information S1.

#### 3.2. Aerosol Particle Number Concentration and Cloud Microphysical Properties

Looking at the  $N_a$ - $R_{eff}$  and  $N_a$ -CDNC linear relationships reported in Figures 2a and 2b respectively,  $R_{eff}$  decreases with increasing  $N_a$  concentration while CDNC naturally increases with increasing  $N_a$ , which is consistent with the well-established Twomey effect (Twomey, 1974). Despite an apparent data scattering, likely arising from comparing ground measurements with ground remote sensing retrievals as well as from differences in other controlling factors such as updraft velocity, aerosol size distributions, and aerosol composition, the two regressions are statistically significant (p < 0.05). From the linear regression analysis, it can be inferred that the variations in aerosol concentration describe up to 21% and 36% of the  $R_{eff}$  and CDNC variance, respectively.

We quantified the cloud microphysical response to aerosol loading by using the quantitative ACI index (the slope of the relationship between aerosol loading and cloud parameter in log-log space). At constant LWC, the ACI index evaluates the susceptibility of  $R_{eff}$  and CDNC to changes in aerosol number concentration, by varying between 0 (no effect at all) to 0.33 (linear response). The  $N_a$ - $R_{eff}$  and  $N_a$ -CDNC relationships, grouped for homogeneous LWC classes, are presented in Figures 2c and 2d, respectively. Both ACI<sub>r</sub> and ACI<sub>n</sub> from the three LWC bins are positive, implying that clouds are sensitive to aerosol number concentration regardless of the amount of liquid water in the atmosphere (the differences between slopes are non-significant at p < 0.05). The range of ACI<sub>r</sub> values observed (0.11–0.16) is consistent with previous findings using ground-based measurements. For instance, ACI<sub>r</sub> values ranged from 0.04 to 0.17 (Kim et al., 2008), using a triennial analysis (1999-2001). The ACI<sub>r</sub> values between 0.02 and 0.19 were derived from an intensive operation period during May 2003 at a continental site of the United States (Feingold et al., 2003) and around 0.10–0.19 based on in situ aircraft observations in September 2015 over Hebei, China (Zhao et al., 2018). By considering in situ ground-based remote sensing for stratiform clouds, the ACI<sub>r</sub> values ranged from 0.05 to 0.16 on the California coast (McComiskey et al., 2009) and from 0.13 to 0.19 in the Arctic regions (Garrett et al., 2004). In addition, the ACI<sub>n</sub> ranges from 0.2 to 0.22 in line with what was found in clean air Arctic summer conditions (Koike et al., 2019).

In summary, the microphysical properties of marine stratiform clouds advected over the NEA Ocean readily respond to changes in aerosol number concentration below the cloud base. This demonstrates that marine aerosols may significantly impact the microphysical properties of marine clouds over the NEA Ocean being an integral part of the oceanic biota-cloud-climate feedback system.

#### 3.3. Aerosol Impact on Cloud Optical Properties Mediated by CDNC

It is well established that cloud albedo and COD in liquid clouds are primarily a function of liquid water path (Frey et al., 2017), therefore any modifications of cloud optical properties attributable to aerosol-induced microphysical





**Figure 2.** (a) Scatter plot between  $N_a$  and  $R_{eff}$  (b) Scatter plot between  $N_a$  and cloud droplet number concentration (CDNC). (c) and (d) aerosol-cloud interaction (ACI)<sub>r</sub> derived from  $R_{eff}$  to  $N_a$  and ACI<sub>n</sub> derived from CDNC to  $N_a$  at three liquid water content (LWC) bins classified as LWC < third percentile (blue), one-third percentile < LWC < two-third percentile (green) and LWC > two-third percentile (red); see Section 2.6 for the motivation of selecting homogeneous LWC classes in calculating ACI. All regression lines are significant (p < 0.05) whereas the differences between the slopes at different LWC classes are non-significant (p < 0.05).

perturbations must be evaluated along with possible adjustments of cloud macrophysical properties. We have studied the relationship of cloud geometrical parameters ( $H_{base}$ ,  $H_{top}$  and  $H_{thick}$ ), LWC, albedo, and COD retrieved at MHD by the SYRSOC algorithm with  $N_a$  and CHL, along with the main meteorological parameters controlling the occurrence of stratiform clouds in this area of the world ocean (Table 1). The ECMWF-ERA5 accessed from the Copernicus Climate Change Service was used to perform spatial correlation analysis, similar to that done with CHL. No time-lag effect on the correlations between cloud parameters and meteorological variables was found. The resulting correlation maps are shown in Figures S12-S16 in Supporting Information S1. The maps



#### Table 1

Correlation Coefficients Between Clean Marine Cloud Data Measured at MHD and Meteorological Variables

Variable	Т	SST	RH	PVV	LTS	CHL	N <sub>a</sub>
R <sub>eff</sub>	-0.33	-0.28				-0.46	-0.50
CDNC	0.27	0.22			0.27	0.44	0.65
H <sub>base</sub>			-0.72		0.75	0.23	0.41
H <sub>top</sub>		0.20	-0.71		0.74	0.21	0.38
H <sub>thick</sub>			0.25	-0.32	-0.24		-0.23
Albedo			0.50		-0.37		
Albedol			0.55		-0.30		
COD			0.44		-0.29		
<b>ICODI</b>			0.54		-0.29		

*Note.* Correlations with CHL and N<sub>a</sub> were added for comparison. Coefficients reported in bold are statistically significant at p < 0.05, while those in italics are statistically significant at p < 0.1. The sign || indicates data normalized to cloud thickness. Coefficients associated with p > 0.2 have not been reported.

show that correlations, when present, are evident only in a region close to MHD ( $50^{\circ}-55^{\circ}N$  and  $10^{\circ}-15^{\circ}W$ ; dashed box in the presented maps, hereafter referred to by MHD box) and do not extend over the oceanic region identified by the relationship with CHL.

It can be inferred from Table 1 that cloud microphysical properties ( $R_{eff}$  and CDNC) are mainly controlled by CHL (proxy of the oceanic biological activity) and aerosol number concentration, with a minor association to T and/or SST (which both tend to co-vary with marine biological activity). Conversely, Cloud macrophysical ( $H_{base}$ ,  $H_{top}$ , and  $H_{thick}$ ) and optical properties (COD and albedo) are mainly influenced by meteorological conditions such as RH and LTS, as expected, with no apparent linear dependence on either temperature or biological activity. Increased RH (more water available in the atmosphere) is associated with lower  $H_{base}$  and  $H_{top}$  values, due to rapid activation of cloud nuclei, as well as to enhanced cloud albedo and COD. The parameters most influenced by LTS are  $H_{base}$  and  $H_{top}$ , which tend to increase in unstable conditions (deeper boundary layer), in agreement with the results from both observational (Wood & Bretherton, 2006) and modeling (Bretherton et al., 2013) studies.

CHL and  $\mathrm{N}_{\mathrm{a}}$  do not show significant linear correlations with the SYRSOC

retrieved cloud albedo, nor with the normalized albedo (albedo/cloud thickness). This suggests that, although there is a noticeable effect on the cloud microphysical properties, the emissions from the marine biota are not linearly related to cloud reflectivity. On the other hand, the effect of cloud microphysics on the normalized optical properties (albedo and COD) is significant (p < 0.05; Figure S17 in Supporting Information S1). The correlation coefficients between CDNC and normalized albedo (normalized COD) are equal to 0.50 (0.43). Our data show that the most important microphysical parameter in controlling cloud albedo is LWC, in agreement with previous studies (Frey et al., 2017; Liu et al., 2020). LWC accounts for up to 53% (R = 0.73; slope = 0.85; intercept = 0.06) of the albedo variance (Figure S18 in Supporting Information \$1). Past studies have shown that aerosol effects on cloud albedo are mediated through a chain of cloud microphysical and macrophysical properties, all linked by two-way feedback. Recently a summary of potential (and sometimes contrasting) mechanisms relevant to marine stratocumulus clouds were provided by Glassmeier et al. (2021). Depending on relative humidity, the occurrence of precipitations, entrainment rate, and CDNC concentrations, the sensitivity of cloud albedo to aerosol perturbations can vary from positive to negative. As a result, for the same geographical area, both positive and negative sensitivities to CDNC can be found. Such sensitivities can be quantified, on a statistical basis, using the joint probability histograms approach (Gryspeerdt et al., 2019). We utilized the joint histograms data analysis approach on the SYRSOC dataset from MHD to investigate the dependence of LWC and albedo on CDNC. The use of joint histograms is preferable in this case, with respect to a simple linear regression, as it can evidence also nonlinear relationships between CDNC and LWC or albedo and account for dispersion of the data caused by the variability in the boundary (meteorological) conditions, hence providing more accurate information about the shape of the relationship. This approach has been previously used to analyze the dependence of cloud fraction (Gryspeerdt et al., 2016) and albedo (Gryspeerdt et al., 2017) on aerosol properties.

The CDNC-LWC and CDNC-albedo joint histograms (Figure 3) and the log-log linear fit lines show that such relationships are mostly non-linear and positive. An increase in the LWC and albedo with increasing CDNC is observed particularly in the bin range of  $20-200 \text{ cm}^{-3}$  (low CDNC) and in the range of LWC from 0.1 to 0.5 g m<sup>-3</sup>, suggesting that increases of marine biogenic aerosol concentrations lead, on average, to LWC and albedo enhancements in stratiform clouds over the NEA Ocean. Positive adjustments of cloud LWC to aerosol perturbations are commonly explained by increased colloidal stability of clouds of reduced droplet size with a consequent effect of precipitation suppression (see, the discussion below). It should be noted, however, that joint histograms only provide a statistical representation of the relationship between cloud properties and aerosols and only allow us to infer the possible mechanisms involved in such a relationship. This is the first application of joint histograms to the analysis of susceptibility of cloud properties to aerosols on the basis of ground-based cloud retrievals. With aim of comparison with analogous representations based on global satellite observations, the analysis of



# Journal of Geophysical Research: Atmospheres



Figure 3. Joint probability histograms of (a) cloud droplet number concentration – liquid water content (CDNC-LWC) and (b) CDNC-Albedo; darker colors indicate higher probability; the colors are normalized so that the sum of the pixels is one in each plot. The blue line is the linear regression fit on the log-log data and the linear sensitivity is inserted at the top of each plot. Sensitivities were calculated as the slope of the linear regression coefficient in the log-log scale.

Gryspeerdt et al. (2019) also shows positive adjustments of liquid water path to CDNC perturbations in the low concentrations range (10–30 cm<sup>-3</sup>) turning positive at higher CDNC. Our study suggests that in the NEA Ocean, positive adjustments can be associated with CDNC changes extending beyond 100 cm<sup>-3</sup>, hence to a wider range compared to previous reports.

## 4. Discussion

By using a unique dataset of multi-year (2009-2015) remote sensing cloud observations, the effect of the oceanic biological activity on cloud properties was investigated over the NEA Ocean. The retrieved cloud microphysical parameters fall in the same range as those obtained from satellite (MODIS) observations (Figure S2 in Supporting Information S1) – widely used in previous studies – but do not require assumptions of adiabaticity and are much less affected by common retrieval errors affecting satellite cloud observations (e.g., from the solar zenith angle; Gryspeerdt et al. [2019]). This allows us to investigate the subtle ACI effects on cloud properties at the low aerosol concentration domain typical of marine background conditions.

Our data show that the oceanic biota affects marine cloud microphysical properties by affecting the aerosol number concentration in the MBL. The effect is a reduction of  $R_{eff}$  coupled to enhanced CDNC during periods of HBA, compared to the winter baseline period. The ACI index shows that marine stratiform clouds, over the NEA, are sensitive to aerosol number concentration variations supporting the above findings. These results are consistent with the recent findings by Rosenfeld et al. (2019), which concluded that the variability in aerosol concentration can explain ~45% of the variability of the cloud radiative effect of oceanic low-level clouds over the Southern Ocean, using satellite observations. Within the range of variability encountered in our dataset, ranging from LBA to HBA conditions (winter to summer) induces an increase in surface CHL of ~2.4 times in seawaters facing MHD, associated with a doubling in CDNC and a decrease of  $R_{eff}$  by ~14% (Figure 4).

Quantifying the aerosol effects on cloud albedo is less straightforward due to the complexity of the relationship between cloud reflectivity and microphysical properties. The LWC sensitivity to CDNC is important since it may enhance or offset the overall albedo sensitivity (Glassmeier et al., 2021). The observed positive adjustments of stratiform cloud macrophysics to aerosol perturbations can be partly explained by precipitation suppression effects (Albrecht, 1989). Smaller, more numerous droplets can diminish collision coalescence and delay precipitation onset (Christensen et al., 2020); longer-lived clouds have a stronger cooling effect on climate. Precipitation is considered as strongly suppressed when  $R_{eff}$  does not exceed 12–14 µm in marine low clouds (Fan et al., 2020; Freud & Rosenfeld, 2012). In our study, the  $R_{eff}$  was below 12 µm in 54% of the time steps (4,259 of 7,884 data



# Journal of Geophysical Research: Atmospheres



**Figure 4.** (a) The direction of background marine and polluted continental air masses arriving at Mace Head station. (b) Seasonal average sea surface chlorophyll a concentration during 2009-2015, illustrating low biological activity in the Northeast Atlantic waters during winter (LBA) and high activity in summer (HBA). (c) Quantification of change in cloud microphysical properties passing from Polluted to Clean marine air masses. (d) Quantification of change in cloud microphysical properties passing from LBA to HBA.

points) during LBA and in 72% of the time steps (24,230 of 33,808 data points) during the HBA. The actual role played by aerosol-induced precipitation suppression associated with marine biogenic emissions in the NEA deserves further studies.

The recent literature focusing on cloud adjustments to marine aerosol perturbations provides a wide range (from negative to positive) sensitivity. A positive LWC sensitivity is more common at small CDNC values, while the sign turns negative above a certain threshold which depends on the cloud system and on meteorological constraints (Glassmeier et al., 2021). As expected, and in agreement with the results of satellite observations of marine clouds in the Southern Ocean (Rosenfeld et al., 2019) and the Northwest Atlantic Ocean (Sinclair et al., 2020), our findings show that the variability of cloud macrophysical properties is primarily dictated by meteorological conditions (Table 1). Nevertheless, our analysis demonstrates a positive non-linear relationship between CDNC and both cloud LWC and albedo (Figure 3). In consistency with our findings, Michibata et al. (2016) showed a positive sensitivity of liquid water path to CDNC over the NEA Ocean employing satellite data (A-Train). However, satellite observations (Chen et al., 2014; Gryspeerdt et al., 2019) and global model simulations (Lebo & Feingold, 2014; Sato et al., 2018) reported a negative response of liquid water path to CDNC, as a result of enhanced evaporation-entrainment feedbacks (Ackerman et al., 2004; Wang et al., 2003; Xue et al., 2008), which unreels to some extent the cloud albedo effect. In spite of such a large span in marine cloud susceptibilities to aerosol perturbations, most studies relying on global satellite observations consistently report more negative adjustments in the tropics with respect to the mid-latitudes (Gryspeerdt et al., 2019; Michibata et al., 2016). The actual range of aerosol concentrations relevant for actual positive cloud adjustments is a matter of debate. Our study suggests that perturbations in CDNC over a wide range, from 20 to 200 cm<sup>-3</sup> propagate to positive adjustments in stratus cloud LWC and albedo in the NEA, hence supporting the importance of mechanisms associated with precipitation suppression.

Our study supports the importance of aerosol-cloud interactions in the marine atmosphere triggered by the emissions from the marine biota. Variations in cloud microphysical properties associated with marine aerosol emissions and similar to the ones observed in this study have been associated with significant radiative effects in the



literature. Observations in the Southern Ocean, typically considered a unique clean marine environment for its remoteness, showed that ocean biology can double the CDNC over biologically productive waters in summer, with respect to unproductive waters (McCoy et al., 2015). It was calculated that such a change in CDNC can cause a net reflection in shortwave radiation between -4 and -6 W m<sup>-2</sup> as an annual average, exceeding -10 W m<sup>-2</sup> in summer. A similar doubling in CDNC has been reported (Meskhidze & Nenes, 2006) over a blooming area of the Southern Ocean with respect to non-blooming regions, associated with a more intense reduction of  $R_{eff}$  (~30%); such variation in cloud microphysical properties was estimated to cause a strong cooling reaching -15 W m<sup>-2</sup>. The aforementioned values are roughly equivalent to the yearly mean radiative forcing estimated in global circulation models as a result of aerosol-cloud interaction over severely polluted regions (Bian & Prather, 2002; Jones et al., 1994; Menon et al., 2002; Rap et al., 2013; Zelinka et al., 2014). Further studies in the Southern Ocean quantified the effect of secondary sulfate emissions on the annual mean reflected shortwave radiation to be -9.32 W m<sup>-2</sup> (Thomas et al., 2010), as a result of a 15%–18% reduction in  $R_{eff}$ . To this same seasonal cycle of  $R_{eff}$  over the Southern Ocean, McCoy et al. (2014) attributed a difference in the reflection of the shortwave radiation of up to -8 W m<sup>-2</sup>.

As observed from the MHD location, the NEA can be considered pristine 64% of the time (Grigas et al., 2017). In other model simulation studies, the occurrence of pristine air masses over the NEA is estimated to be even lower (Hamilton et al., 2014). Consequently, one can argue that anthropogenic aerosol particles may have a much stronger impact on cloud microphysical properties than biogenic ones. By comparing background marine conditions to anthropogenically influenced cases (Figure 4 and Table S2 in Supporting Information S1), we observed an enhancement of CDNC by 3 times and a 25% reduction in  $R_{eff}$ , passing from unperturbed marine clouds to polluted conditions. These variations are not far from those observed passing from background LBA to HBA conditions, doubling in CDNC and decreasing in  $R_{eff}$  by ~14% (Figure 4). These findings suggest that, over the NEA ocean, the impact of oceanic biological activity on cloud properties may be of similar magnitude to that exerted by anthropogenic and continental aerosols and, in any case, not negligible in comparison.

## 5. Conclusions

This work investigates the impact of oceanic biological activity on the properties of stratiform clouds over the NEA using cloud remote sensing observations at Mace Head from 2009 to 2015. Our analysis indicates the potential of accurate remote-sensed cloud parametefrom ground-based measurements for aerosol-cloud radiative effect studies. The main finding of the present work is that marine biota affects cloud microphysical properties by diminishing (-14%) cloud droplet effective radius ( $R_{eff}$ ) and enhancing (+100%) CDNC during periods of enhanced biological activity with respect to the quiescent wintertime. This occurs because the oceanic biological activity influences the concentration of aerosol particles, in the CCN relevant-size range, in the unperturbed marine boundary layer.

Cloud macrophysical properties are mainly controlled by meteorological constraints (tropospheric stability and water vapor availability), and the variability of cloud albedo is primarily dictated by changes in cloud liquid water content. Anyhow, after normalizing for the meteorological constraints, aerosol perturbations on cloud liquid water content and cloud albedo are detectable, and such response results are positive over a wide range of CDNC, and higher with respect to the outcomes of previous studies based on global analyses. It follows that the phytoplankton-induced enhancement in the CCN concentration, occurring during spring-summer over the NEA Ocean, contributes to generally brighter stratiform clouds.

Variations in cloud microphysical properties similar to the ones observed in this study have been associated with significant radiative effects over the Southern Ocean. This suggests that the net radiative effect of the observed winter-summer variation in cloud properties may not be negligible over the NEA either. Along the same line, the magnitude of the variation in cloud microphysical properties between high and low oceanic biological activity periods is comparable to that between anthropogenically influenced and pristine marine air masses. This implies that, over the NEA, the seasonal CCN fluctuations due to biological activity may have a similar radiative impact as those caused by the advection of anthropogenic continental aerosols.

The uncertainty of climate models associated with aerosol-cloud interactions is largely due to the lack of accurate parameterizations and the limited understanding of the complex feedbacks. Our results demonstrate an evident impact of the oceanic biota on marine stratiform cloud properties over the NEA, where radiative forcing is



traditionally considered anthropogenically driven in climate models. Our study indicates that CCN variability impacted by marine biogenic emissions can be a key factor to constrain to reducing the uncertainty of radiative forcing from aerosol-cloud interactions over the global ocean.

#### **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

#### **Data Availability Statement**

The SYRSOC results for the 118 cloud cases can be accessed at http://doi.org/10.5281/zenodo.154137. Measurement times of the selected 52 background marine cases are shown in Data S1 (supplementary materials). The remote-sensing CHL dataset is available at http://marine.copernicus.eu/ (Identifier: OCEANCOLOUR\_GLO\_ CHL\_L4\_REP\_OBSERVATIONS\_009\_082). The ERA5 meteorological dataset can be accessed at https://cds. climate.copernicus.eu/. The air mass back-trajectories data are archived through https://www.arl.noaa.gov/. The codes used in this study for data elaborations are available from GitHub at https://github.com/karam-mansour/ Phytoplankton\_Cloud\_North\_Atlantic and can be cited through https://doi.org/10.5281/zenodo.6489668.

#### Acknowledgments

The authors would like to thank the EPA Ireland and the Department of the Environment, Climate and Communications: MaREI, the SFI Research Centre for Energy, Climate, and Marine for supporting this research. We gratefully acknowledge the NOAA Air Resources Laboratory (ARL) for provision of the HYSPLIT transport and dispersion model, the EU Copernicus Marine Environment Monitoring Service (CMEMS) for provision satellite ocean color data, the Copernicus climate change service (C3S) for providing ERA5 reanalysis meteorological data, and the Terra and Aqua mission scientists and associated NASA personnel for production and provision of MODIS data. The research was funded by the European Commission: H2020 Research Infrastructures, project FORCeS (Grant No. 821205) and FP7-ENV- 2013 project BACCHUS (Grant No. 603445), Open Access Funding provided by Consiglio Nazionale delle Ricerche within the CRUI-CARE Agreement.

## References

- Ackerman, A. S., Kirkpatrick, M. P., Stevens, D. E., & Toon, O. B. (2004). The impact of humidity above stratiform clouds on indirect aerosol climate forcing. *Nature*, 432(7020), 1014–1017. https://doi.org/10.1038/nature03174
- Albrecht, B. A. (1989). Aerosols, cloud microphysics, and fractional cloudiness. Science, 245(4923), 1227–1230. https://doi.org/10.1126/ science.245.4923.1227
- Andersen, H., Cermak, J., Fuchs, J., Knutti, R., & Lohmann, U. (2017). Understanding the drivers of marine liquid-water cloud occurrence and properties with global observations using neural networks. *Atmospheric Chemistry and Physics*, 17(15), 9535–9546. https://doi.org/10.5194/ acp-17-9535-2017
- Ault, A. P., Moffet, R. C., Baltrusaitis, J., Collins, D. B., Ruppel, M. J., Cuadra-Rodriguez, L. A., et al. (2013). Size-dependent changes in sea spray aerosol composition and properties with different seawater conditions. *Environmental Science & Technology*, 47(11), 5603–5612. https:// doi.org/10.1021/es400416g
- Behrenfeld, M. J., O'Malley, R. T., Boss, E. S., Westberry, T. K., Graff, J. R., Halsey, K. H., et al. (2016). Revaluating ocean warming impacts on global phytoplankton. *Nature Climate Change*, 6(3), 323–330. https://doi.org/10.1038/nclimate2838
- Bian, H. S., & Prather, M. J. (2002). Fast-J2: Accurate simulation of stratospheric photolysis in global chemical models. *Journal of Atmospheric Chemistry*, 41(3), 281–296. https://doi.org/10.1023/a:1014980619462
- Boucher, O., Randall, D., Artaxo, P., Bretherton, C., Feingold, G., Forster, P., et al. (2013). Clouds and aerosols. In T. F. Stocker, D. Qin, G. K. Plattner, M. Tignor, S. K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex, & P. M. Midgley (Eds.), Climate change 2013: The physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press.
- Bretherton, C. S., Blossey, P. N., & Jones, C. R. (2013). Mechanisms of marine low cloud sensitivity to idealized climate perturbations: A single-LES exploration extending the CGILS cases. *Journal of Advances in Modeling Earth Systems*, 5(2), 316–337. https://doi.org/10.1002/ jame.20019
- C3S, Copernicus Climate Change Service. (2017). ERA5: Fifth generation of ECMWF atmospheric reanalyses of the global climate. Copernicus Climate Change Service Climate Data Store (CDS). Retrieved from https://cds.climate.copernicus.eu/cdsapp#!/home
- Carslaw, K. S., Lee, L. A., Reddington, C. L., Pringle, K. J., Rap, A., Forster, P. M., et al. (2013). Large contribution of natural aerosols to uncertainty in indirect forcing. *Nature*, 503(7474), 67–71. https://doi.org/10.1038/nature12674
- Charlson, R. J., Lovelock, J. E., Andreae, M. O., & Warren, S. G. (1987). Oceanic phytoplankton, atmospheric sulfur, cloud albedo and climate. *Nature*, 326(6114), 655–661. https://doi.org/10.1038/326655a0
- Chen, Y. C., Christensen, M. W., Stephens, G. L., & Seinfeld, J. H. (2014). Satellite-based estimate of global aerosol-cloud radiative forcing by marine warm clouds. *Nature Geoscience*, 7(9), 643–646. https://doi.org/10.1038/ngeo2214
- Christensen, M. W., Jones, W. K., & Stier, P. (2020). Aerosols enhance cloud lifetime and brightness along the stratus-to-cumulus transition. Proceedings of the National Academy of Sciences of the United States of America, 117(30), 17591–17598. https://doi.org/10.1073/ pnas.1921231117
- Collins, D. B., Bertram, T. H., Sultana, C. M., Lee, C., Axson, J. L., & Prather, K. A. (2016). Phytoplankton blooms weakly influence the cloud forming ability of sea spray aerosol. *Geophysical Research Letters*, 43(18), 9975–9983. https://doi.org/10.1002/2016gl069922
- Crewell, S., & Lohnert, U. (2003). Accuracy of cloud liquid water path from ground-based microwave radiometry 2. Sensor accuracy and synergy. *Radio Science*, 38(3), 7-1–7-3. https://doi.org/10.1029/2002rs002634
- Facchini, M. C., Decesari, S., Rinaldi, M., Carbone, C., Finessi, E., Mircea, M., et al. (2008). Important source of marine secondary organic aerosol from biogenic Amines. *Environmental Science & Technology*, 42(24), 9116–9121. https://doi.org/10.1021/es8018385
- Falkowski, P. G., Kim, Y., Kolber, Z., Wilson, C., Wirick, C., & Cess, R. (1992). Natural versus anthropogenic factors affecting low-level cloud albedo over the North-Atlantic. *Science*, 256(5061), 1311–1313. https://doi.org/10.1126/science.256.5061.1311
- Fan, C. X., Wang, M. H., Rosenfeld, D., Zhu, Y. N., Liu, J. H., & Chen, B. J. (2020). Strong precipitation suppression by aerosols in marine low clouds. *Geophysical Research Letters*, 47(7), e2019GL086207. https://doi.org/10.1029/2019gl086207
- Feingold, G., Eberhard, W. L., Veron, D. E., & Previdi, M. (2003). First measurements of the Twomey indirect effect using ground-based remote sensors. *Geophysical Research Letters*, 30(6), 1287. https://doi.org/10.1029/2002gl016633

- Fossum, K. N., Ovadnevaite, J., Ceburnis, D., Dall'Osto, M., Marullo, S., Bellacicco, M., et al. (2018). Summertime primary and secondary contributions to Southern Ocean cloud condensation nuclei. *Scientific Reports*, 8, 13844. https://doi.org/10.1038/s41598-018-32047-4
- Fossum, K. N., Ovadnevaite, J., Ceburnis, D., Preißler, J., Snider, J. R., Huang, R. J., et al. (2020). Sea-spray regulates sulfate cloud droplet activation over oceans. *Npj Climate and Atmospheric Science*, 3(1), 14. https://doi.org/10.1038/s41612-020-0116-2
- Freud, E., & Rosenfeld, D. (2012). Linear relation between convective cloud drop number concentration and depth for rain initiation. Journal of Geophysical Research: Atmospheres, 117, D02207. https://doi.org/10.1029/2011jd016457
- Frey, L., Bender, F. A. M., & Svensson, G. (2017). Cloud albedo changes in response to anthropogenic sulfate and non-sulfate aerosol forcings in CMIP5 models. Atmospheric Chemistry and Physics, 17(14), 9145–9162. https://doi.org/10.5194/acp-17-9145-2017
- Friedland, K. D., Record, N. R., Asch, R. G., Kristiansen, T., Saba, V. S., Drinkwater, K. F., et al. (2016). Seasonal phytoplankton blooms in the North Atlantic linked to the overwintering strategies of copepods. *Elementa-Science of the Anthropocene*, 4, 1–19. https://doi.org/10.12952/ journal.elementa.000099
- Fuchs, J., Cermak, J., Andersen, H., Hollmann, R., & Schwarz, K. (2017). On the influence of air mass Origin on low-cloud properties in the Southeast Atlantic. *Journal of Geophysical Research: Atmospheres*, 122(20), 11076–11091. https://doi.org/10.1002/2017jd027184
- Gantt, B., Xu, J., Meskhidze, N., Zhang, Y., Nenes, A., Ghan, S. J., et al. (2012). Global distribution and climate forcing of marine organic aerosol - Part 2: Effects on cloud properties and radiative forcing. *Atmospheric Chemistry and Physics*, 12(14), 6555–6563. https://doi.org/10.5194/ acp-12-6555-2012
- Garrett, T. J., Zhao, C., Dong, X., Mace, G. G., & Hobbs, P. V. (2004). Effects of varying aerosol regimes on low-level Arctic stratus. *Geophysical Research Letters*, 31(17). https://doi.org/10.1029/2004gl019928
- Glassmeier, F., Hoffmann, F., Johnson, J. S., Yamaguchi, T., Carslaw, K. S., & Feingold, G. (2021). Aerosol-cloud-climate cooling overestimated by ship-track data. *Science*, 371(6528), 485. https://doi.org/10.1126/science.abd3980
- Grigas, T., Ovadnevaite, J., Ceburnis, D., Moran, E., McGovern, F. M., Jennings, S. G., & O'Dowd, C. (2017). Sophisticated clean air strategies required to mitigate against particulate organic pollution. *Scientific Reports*, 7, 44737. https://doi.org/10.1038/srep44737
- Grosvenor, D. P., Sourdeval, O., Zuidema, P., Ackerman, A., Alexandrov, M. D., Bennartz, R., et al. (2018). Remote sensing of droplet number concentration in warm clouds: A review of the current state of knowledge and perspectives. *Reviews of Geophysics*, 56(2), 409–453. https:// doi.org/10.1029/2017rg000593
- Gryspeerdt, E., Goren, T., Sourdeval, O., Quaas, J., Mulmenstadt, J., Dipu, S., et al. (2019). Constraining the aerosol influence on cloud liquid water path. Atmospheric Chemistry and Physics, 19(8), 5331–5347. https://doi.org/10.5194/acp-19-5331-2019
- Gryspeerdt, E., Quaas, J., & Bellouin, N. (2016). Constraining the aerosol influence on cloud fraction. Journal of Geophysical Research: Atmospheres, 121(7), 3566–3583. https://doi.org/10.1002/2015jd023744
- Gryspeerdt, E., Quaas, J., Ferrachat, S., Gettelman, A., Ghan, S., Lohmann, U., et al. (2017). Constraining the instantaneous aerosol influence on cloud albedo. Proceedings of the National Academy of Sciences of the United States of America, 114(19), 4899–4904. https://doi.org/10.1073/ pnas.1617765114
- Hamilton, D. S., Lee, L. A., Pringle, K. J., Reddington, C. L., Spracklen, D. V., & Carslaw, K. S. (2014). Occurrence of pristine aerosol environments on a polluted planet. *Proceedings of the National Academy of Sciences of the United States of America*, 111(52), 18466–18471. https:// doi.org/10.1073/pnas.1415440111
- Heese, B., Flentje, H., Althausen, D., Ansmann, A., & Frey, S. (2010). Ceilometer lidar comparison: Backscatter coefficient retrieval and signalto-noise ratio determination. Atmospheric Measurement Techniques, 3(6), 1763–1770. https://doi.org/10.5194/amt-3-1763-2010
- Hendrickson, B. N., Brooks, S. D., Thornton, D. C. O., Moore, R. H., Crosbie, E., Ziemba, L. D., et al. (2021). Role of sea surface microlayer properties in cloud formation. *Frontiers in Marine Science*, 7. https://doi.org/10.3389/fmars.2020.596225
- Huot, Y., Brown, C. A., & Cullen, J. J. (2007). Retrieval of phytoplankton biomass from simultaneous inversion of reflectance, the diffuse attenuation coefficient, and Sun-induced fluorescence in coastal waters. *Journal of Geophysical Research-Oceans*, 112(C6), C06013. https://doi. org/10.1029/2006jc003794
- Illingworth, A. J., Hogan, R. J., O'Connor, E. J., Bouniol, D., Brooks, M. E., Delanoe, J., et al. (2007). Cloudnet continuous evaluation of cloud profiles in seven operational models using ground-based observations. *Bulletin of the American Meteorological Society*, 88(6), 883. https:// doi.org/10.1175/bams-88-6-883
- Janssen, R. H. H., Ganzeveld, L. N., Kabat, P., Kulmala, M., Nieminen, T., & Roebeling, R. A. (2011). Estimating seasonal variations in cloud droplet number concentration over the boreal forest from satellite observations. *Atmospheric Chemistry and Physics*, 11(15), 7701–7713. https://doi.org/10.5194/acp-11-7701-2011
- Jones, A., Roberts, D. L., & Slingo, A. (1994). A climate model study OF indirect radiative forcing BY anthropogenic sulfate aerosols. Nature, 370(6489), 450–453. https://doi.org/10.1038/370450a0
- Kim, B. G., Miller, M. A., Schwartz, S. E., Liu, Y. G., & Min, Q. L. (2008). The role of adiabaticity in the aerosol first indirect effect. Journal of Geophysical Research: Atmospheres, 113(D5), D05210. https://doi.org/10.1029/2007jd008961
- Klein, S. A., & Hartmann, D. L. (1993). The seasonal cycle OF low stratiform clouds. Journal of Climate, 6(8), 1587–1606. https://doi. org/10.1175/1520-0442(1993)006<1587:tscols>2.0.co;2
- Koike, M., Ukita, J., Strom, J., Tunved, P., Shiobara, M., Vitale, V., & Pedersen, C. A. (2019). Year-round in situ measurements of Arctic low-level clouds: Microphysical properties and their relationships with aerosols. *Journal of Geophysical Research: Atmospheres*, 124(3), 1798–1822. https://doi.org/10.1029/2018jd029802

Lacour, L., Claustre, H., Prieur, L., & D'Ortenzio, F. (2015). Phytoplankton biomass cycles in the North Atlantic subpolar gyre: A similar mechanism for two different blooms in the Labrador sea. Geophysical Research Letters, 42(13), 5403–5410. https://doi.org/10.1002/2015gl064540

- Lebo, Z. J., & Feingold, G. (2014). On the relationship between responses in cloud water and precipitation to changes in aerosol. Atmospheric Chemistry and Physics, 14(21), 11817–11831. https://doi.org/10.5194/acp-14-11817-2014
- Liu, Z., Wang, M., Rosenfeld, D., Zhu, Y., Bai, H., Cao, Y., & Liang, Y. (2020). Evaluation of cloud and precipitation response to aerosols in WRF-Chem with satellite observations. *Journal of Geophysical Research: Atmospheres*, 125(18), e33108. https://doi. org/10.1029/2020JD033108
- Loeb, N. G., & Schuster, G. L. (2008). An observational study of the relationship between cloud, aerosol and meteorology in broken low-level cloud conditions. Journal of Geophysical Research: Atmospheres, 113(D14), D14214. https://doi.org/10.1029/2007jd009763
- Lohnert, U., & Crewell, S. (2003). Accuracy of cloud liquid water path from ground-based microwave radiometry 1. Dependency on cloud model statistics. *Radio Science*, 38(3), 8041. https://doi.org/10.1029/2002rs002654
- Lohnert, U., Turner, D. D., & Crewell, S. (2009). Ground-based temperature and humidity profiling using Spectral Infrared and microwave observations. Part I: Simulated retrieval performance in clear-sky conditions. Journal of Applied Meteorology and Climatology, 48(5), 1017–1032. https://doi.org/10.1175/2008jamc2060.1



- Mansour, K., Decesari, S., Bellacicco, M., Marullo, S., Santoleri, R., Bonasoni, P., et al. (2020). Particulate methanesulfonic acid over the central Mediterranean Sea: Source region identification and relationship with phytoplankton activity. *Atmospheric Research*, 237, 104837. https://doi. org/10.1016/j.atmosres.2019.104837
- Mansour, K., Decesari, S., Facchini, M. C., Belosi, F., Paglione, M., Sandrini, S., et al. (2020). Linking marine biological activity to aerosol chemical composition and cloud-relevant properties over the North Atlantic Ocean. *Journal of Geophysical Research: Atmospheres*, 125(13), e32246. https://doi.org/10.1029/2019jd032246
- Martucci, G., Milroy, C., Bower, K., Gallagher, M., Lloyd, G., & O'Dowd, C. D. (2013). Comparison of in-situ, satellite and ground-based remote sensing retrievals of liquid cloud microphysics during MACLOUD. In *Paper presented at the 19th International Conference on Nucleation* and Atmospheric Aerosols (ICNAA). Colorado State University Center for the Arts.
- Martucci, G., Milroy, C., & O'Dowd, C. D. (2010). Detection of cloud-base height using Jenoptik CHM15K and Vaisala CL31 ceilometers. Journal of Atmospheric and Oceanic Technology, 27(2), 305–318. https://doi.org/10.1175/2009jtecha1326.1
- Martucci, G., & O'Dowd, C. D. (2011). Ground-based retrieval of continental and marine warm cloud microphysics. Atmospheric Measurement Techniques, 4(12), 2749–2765. https://doi.org/10.5194/amt-4-2749-2011
- Mayer, K. J., Wang, X. F., Santander, M. V., Mitts, B. A., Sauer, J. S., Sultana, C. M., et al. (2020). Secondary marine aerosol plays a dominant role over primary sea spray aerosol in cloud formation. ACS Central Science, 6(12), 2259–2266. https://doi.org/10.1021/acscentsci.0c00793
- McComiskey, A., & Feingold, G. (2012). The scale problem in quantifying aerosol indirect effects. Atmospheric Chemistry and Physics, 12(2), 1031–1049. https://doi.org/10.5194/acp-12-1031-2012
- McComiskey, A., Feingold, G., Frisch, A. S., Turner, D. D., Miller, M. A., Chiu, J. C., et al. (2009). An assessment of aerosol-cloud interactions in marine stratus clouds based on surface remote sensing. *Journal of Geophysical Research: Atmospheres*, 114, D09203. https://doi. org/10.1029/2008jd011006
- McCoy, D. T., Burrows, S. M., Wood, R., Grosvenor, D. P., Elliott, S. M., Ma, P. L., et al. (2015). Natural aerosols explain seasonal and spatial patterns of Southern Ocean cloud albedo. *Science Advances*, 1(6), e150057. https://doi.org/10.1126/sciady.1500157
- McCoy, D. T., Hartmann, D. L., & Grosvenor, D. P. (2014). Observed Southern Ocean cloud properties and shortwave reflection. Part I: Calculation of SW flux from observed cloud properties. *Journal of Climate*, 27(23), 8836–8857. https://doi.org/10.1175/jcli-d-14-00287.1
- Melchionna, S., Bauer, M., & Peters, G. (2008). A new algorithm for the extraction of cloud parameters using multipeak analysis of cloud radar data - first application and preliminary results. *Meteorologische Zeitschrift*, 17(5), 613–620. https://doi.org/10.1127/0941-2948/2008/0322
- Menon, S., Del Genio, A. D., Koch, D., & Tselioudis, G. (2002). GCM Simulations of the aerosol indirect effect: Sensitivity to cloud parameterization and aerosol burden. *Journal of the Atmospheric Sciences*, 59(3), 692–713. https://doi.org/10.1175/1520-0469(2002)059<0692:gsotai>2.0.co;2
- Meskhidze, N., & Nenes, A. (2006). Phytoplankton and cloudiness in the Southern Ocean. *Science*, *314*(5804), 1419–1423. https://doi.org/10.1126/science.1131779
- Meskhidze, N., & Nenes, A. (2010). Effects of ocean ecosystem on marine aerosol-cloud interaction. Advances in Meteorology, 2010, 239808. https://doi.org/10.1155/2010/239808
- Michibata, T., Suzuki, K., Sato, Y., & Takemura, T. (2016). The source of discrepancies in aerosol-cloud-precipitation interactions between GCM and A-Train retrievals. Atmospheric Chemistry and Physics, 16(23), 15413–15424. https://doi.org/10.5194/acp-16-15413-2016
- Miles, N. L., Verlinde, J., & Clothiaux, E. E. (2000). Cloud droplet size distributions in low-level stratiform clouds. *Journal of the Atmospheric Sciences*, 57(2), 295–311. https://doi.org/10.1175/1520-0469(2000)057<0295:cdsdil>2.0.co;2
- Miller, M. A., & Yuter, S. E. (2008). Lack of correlation between chlorophyll a and cloud droplet effective radius in shallow marine clouds. Geophysical Research Letters, 35(13), L13807. https://doi.org/10.1029/2008gl034354
- Miyazaki, Y., Suzuki, K., Tachibana, E., Yamashita, Y., Müller, A., Kawana, K., & Nishioka, J. (2020). New index of organic mass enrichment in sea spray aerosols linked with senescent status in marine phytoplankton. *Scientific Reports*, 10, 17042. https://doi.org/10.1038/ s41598-020-73718-5
- O'Dowd, C., Ceburnis, D., Ovadnevaite, J., Bialek, J., Stengel, D. B., Zacharias, M., et al. (2015). Connecting marine productivity to sea-spray via nanoscale biological processes: Phytoplankton Dance or Death Disco? *Scientific Reports*, 5, 14883. https://doi.org/10.1038/srep14883
- O'Dowd, C., Ceburnis, D., Ovadnevaite, J., Vaishya, A., Rinaldi, M., & Facchini, M. C. (2014). Do anthropogenic, continental or coastal aerosol sources impact on a marine aerosol signature at Mace Head? *Atmospheric Chemistry and Physics*, 14(19), 10687–10704. https://doi. org/10.5194/acp-14-10687-2014
- O'Dowd, C. D., Facchini, M. C., Cavalli, F., Ceburnis, D., Mircea, M., Decesari, S., et al. (2004). Biogenically driven organic contribution to marine aerosol. *Nature*, 431(7009), 676–680. https://doi.org/10.1038/nature02959
- Ovadnevaite, J., Ceburnis, D., Martucci, G., Bialek, J., Monahan, C., Rinaldi, M., et al. (2011). Primary marine organic aerosol: A dichotomy of low hygroscopicity and high CCN activity. *Geophysical Research Letters*, 38, L21806. https://doi.org/10.1029/2011g1048869
- Ovadnevaite, J., Zuend, A., Laaksonen, A., Sanchez, K. J., Roberts, G., Ceburnis, D., et al. (2017). Surface tension prevails over solute effect in organic-influenced cloud droplet activation. *Nature*, 546(7660), 637–641. https://doi.org/10.1038/nature22806
- Painemal, D. (2018). Global estimates of changes in shortwave low-cloud albedo and Fluxes due to variations in cloud droplet number concentration derived from CERES-MODIS satellite sensors. *Geophysical Research Letters*, 45(17), 9288–9296. https://doi.org/10.1029/2018gl078880
- Platnick, S., Meyer, K. G., King, M. D., Wind, G., Amarasinghe, N., Marchant, B., et al. (2017). The MODIS cloud optical and microphysical products: Collection 6 updates and examples from Terra and Aqua. *IEEE Transactions on Geoscience and Remote Sensing*, 55(1), 502–525. https://doi.org/10.1109/tgrs.2016.2610522
- Prather, K. A., Bertram, T. H., Grassian, V. H., Deane, G. B., Stokes, M. D., DeMott, P. J., et al. (2013). Bringing the ocean into the laboratory to probe the chemical complexity of sea spray aerosol. *Proceedings of the National Academy of Sciences of the United States of America*, 110(19), 7550–7555. https://doi.org/10.1073/pnas.1300262110
- Preißler, J. (2016). Dataset: Six years of surface remote sensing of stratiform warm clouds in marine and continental air over Mace Head, Ireland [Data set]. https://doi.org/10.5281/zenodo.154137
- Preißler, J., Martucci, G., Saponaro, G., Ovadnevaite, J., Vaishya, A., Kolmonen, P., et al. (2016). Six years of surface remote sensing of stratiform warm clouds in marine and continental air over Mace Head, Ireland. *Journal of Geophysical Research: Atmospheres*, 121(24), 14538–14557. https://doi.org/10.1002/2016jd025360
- Quinn, P. K., & Bates, T. S. (2011). The case against climate regulation via oceanic phytoplankton sulphur emissions. *Nature*, 480(7375), 51–56. https://doi.org/10.1038/nature10580
- Quinn, P. K., Bates, T. S., Coffman, D. J., Upchurch, L., Johnson, J. E., Moore, R., et al. (2019). Seasonal variations in Western North Atlantic remote marine aerosol properties. *Journal of Geophysical Research: Atmospheres*, 124(24), 14240–14261. https://doi.org/10.1029/2019jd031740
- Rap, A., Scott, C. E., Spracklen, D. V., Bellouin, N., Forster, P. M., Carslaw, K. S., et al. (2013). Natural aerosol direct and indirect radiative effects. *Geophysical Research Letters*, 40(12), 3297–3301. https://doi.org/10.1002/grl.50441

- Rinaldi, M., Fuzzi, S., Decesari, S., Marullo, S., Santoleri, R., Provenzale, A., et al. (2013). Is chlorophyll-a the best surrogate for organic matter enrichment in submicron primary marine aerosol? *Journal of Geophysical Research: Atmospheres*, 118(10), 4964–4973. https://doi. org/10.1002/jgrd.50417
- Rinaldi, M., Hiranuma, N., Santachiara, G., Mazzola, M., Mansour, K., Paglione, M., et al. (2021). Ice-nucleating particle concentration measurements from Ny-Alesund during the Arctic spring-summer in 2018. Atmospheric Chemistry and Physics, 21(19), 14725–14748. https://doi. org/10.5194/acp-21-14725-2021
- Rolph, G., Stein, A., & Stunder, B. (2017). Real-time environmental applications and Display system: Ready. Environmental Modelling & Software, 95, 210–228. https://doi.org/10.1016/j.envsoft.2017.06.025
- Rosenfeld, D., Zhu, Y. N., Wang, M. H., Zheng, Y. T., Goren, T., & Yu, S. C. (2019). Aerosol-driven droplet concentrations dominate coverage and water of oceanic low-level clouds. *Science*, 363(6427), 599. https://doi.org/10.1126/science.aav0566
- Saliba, G., Chen, C. L., Lewis, S., Russell, L. M., Quinn, P. K., Bates, T. S., et al. (2020). Seasonal differences and variability of concentrations, chemical composition, and cloud condensation nuclei of marine aerosol over the North Atlantic. *Journal of Geophysical Research: Atmos*pheres, 125(19), e2020JD033145. https://doi.org/10.1029/2020jd033145
- Sanchez, K. J., Chen, C. L., Russell, L. M., Betha, R., Liu, J., Price, D. J., et al. (2018). Substantial seasonal contribution of observed biogenic sulfate particles to cloud condensation nuclei. *Scientific Reports*, 8, 3235. https://doi.org/10.1038/s41598-018-21590-9
- Sanchez, K. J., Roberts, G. C., Calmer, R., Nicoll, K., Hashimshoni, E., Rosenfeld, D., et al. (2017). Top-down and bottom-up aerosol-cloud closure: Towards understanding sources of uncertainty in deriving cloud shortwave radiative flux. *Atmospheric Chemistry and Physics*, 17(16), 9797–9814. https://doi.org/10.5194/acp-17-9797-2017
- Satheesh, S. K., & Moorthy, K. K. (2005). Radiative effects of natural aerosols: A review. Atmospheric Environment, 39(11), 2089–2110. https://doi.org/10.1016/j.atmosenv.2004.12.029
- Sato, Y., Goto, D., Michibata, T., Suzuki, K., Takemura, T., Tomita, H., & Nakajima, T. (2018). Aerosol effects on cloud water amounts were successfully simulated by a global cloud-system resolving model. *Nature Communications*, 9, 985. https://doi.org/10.1038/s41467-018-03379-6 uccessfully simulated by a global cloud-system resolving model. *Nature Communications*, 9, 985. https://doi.org/10.1038/s41467-018-03379-6
- Sato, Y., & Suzuki, K. (2019). How do aerosols affect cloudiness? *Science*, *363*(6427), 580–581. https://doi.org/10.1126/science.aaw3720 Sellegri, K., Nicosia, A., Freney, E., Uitz, J., Thyssen, M., Gregori, G., et al. (2021). Surface ocean microbiota determine cloud precursors. *Scien-*
- tific Reports, 11(1), 281. https://doi.org/10.1038/s41598-020-78097-5 Sinclair, K., van Diedenhoven, B., Cairns, B., Alexandrov, M., Moore, R., Ziemba, L. D., & Crosbie, E. (2020). Observations of aerosol-cloud interactions during the North Atlantic aerosol and marine ecosystem study. *Geophysical Research Letters*, 47(3), e85851. https://doi.org/10.1029/2019g1085851
- Sorooshian, A., & Duong, H. T. (2010). Ocean emission effects on aerosol-cloud interactions: Insights from two case studies. Advances in Meteorology, 2010, 301395. https://doi.org/10.1155/2010/301395
- Sorooshian, A., Padro, L. T., Nenes, A., Feingold, G., McComiskey, A., Hersey, S. P., et al. (2009). On the link between ocean biota emissions, aerosol, and maritime clouds: Airborne, ground, and satellite measurements off the coast of California. *Global Biogeochemical Cycles*, 23, GB4007. https://doi.org/10.1029/2009gb003464
- Stein, A. F., Draxler, R. R., Rolph, G. D., Stunder, B. J. B., Cohen, M. D., & Ngan, F. (2015). NOAA'S HYSPLIT atmospheric transport and dispersion modeling system. Bulletin of the American Meteorological Society, 96(12), 2059–2077. https://doi.org/10.1175/bams-d-14-00110.1
- Stevens, B., & Feingold, G. (2009). Untangling aerosol effects on clouds and precipitation in a buffered system. *Nature*, 461(7264), 607–613. https://doi.org/10.1038/nature08281
- Thomas, M. A., Suntharalingam, P., Pozzoli, L., Rast, S., Devasthale, A., Kloster, S., et al. (2010). Quantification of DMS aerosol-cloud-climate interactions using the ECHAM5-HAMMOZ model in a current climate scenario. Atmospheric Chemistry and Physics, 10(15), 7425–7438. https://doi.org/10.5194/acp-10-7425-2010
- Twomey, S. (1974). Pollution and the planetary albedo. *Atmospheric Environment*, 8(12), 1251–1256. https://doi.org/10.1016/0004-6981(74)90004-3 Wang, S. P., Wang, Q., & Feingold, G. (2003). Turbulence, condensation, and liquid water transport in numerically simulated nonprecipitating stra-
- tocumulus clouds. Journal of the Atmospheric Sciences, 60(2), 262–278. https://doi.org/10.1175/1520-0469(2003)060<0262:tcalwt>2.0.co;2 Wood, R., & Bretherton, C. S. (2006). On the relationship between stratiform low cloud cover and lower-tropospheric stability. Journal of
- Climate, 19(24), 6425–6432. https://doi.org/10.1175/jcli3988.1 Xue, H. W., Feingold, G., & Stevens, B. (2008). Aerosol effects on clouds, precipitation, and the organization of shallow cumulus convection.
- Journal of the Atmospheric Sciences, 65(2), 392–406. https://doi.org/10.1175/2007jas2428.1 Yoon, Y. J., Ceburnis, D., Cavalli, F., Jourdan, O., Putaud, J. P., Facchini, M. C., et al. (2007). Seasonal characteristics of the physicochemical properties of North Atlantic marine atmospheric aerosols. Journal of Geophysical Research: Atmospheres, 112(D4), D04206. https://doi. org/10.1029/2005id007044
- Zelinka, M. D., Andrews, T., Forster, P. M., & Taylor, K. E. (2014). Quantifying components of aerosol-cloud-radiation interactions in climate models. Journal of Geophysical Research: Atmospheres, 119(12), 7599–7615. https://doi.org/10.1002/2014jd021710

Zhao, C. F., Klein, S. A., Xie, S. C., Liu, X. H., Boyle, J. S., & Zhang, Y. Y. (2012). Aerosol first indirect effects on non-precipitating low-level liquid cloud properties as simulated by CAM5 at ARM sites. *Geophysical Research Letters*, *39*, L08806. https://doi.org/10.1029/2012gl051213

Zhao, C. F., Qiu, Y. M., Dong, X. B., Wang, Z. E., Peng, Y. R., Li, B. D., et al. (2018). Negative Aerosol-Cloud r(e) Relationship From Aircraft Observations Over Hebei, China. *Earth and Space Science*, 5(1), 19–29. https://doi.org/10.1002/2017ea000346

## **References From the Supporting Information**

- Brenguier, J. L., Pawlowska, H., Schuller, L., Preusker, R., Fischer, J., & Fouquart, Y. (2000). Radiative properties of boundary layer clouds: Droplet effective radius versus number concentration. *Journal of the Atmospheric Sciences*, 57(6), 803–821. https://doi. org/10.1175/1520-0469(2000)057<0803:rpoblc>2.0.co;2
- Grosvenor, D. P., & Wood, R. (2014). The effect of solar zenith angle on MODIS cloud optical and microphysical retrievals within marine liquid water clouds. *Atmospheric Chemistry and Physics*, 14(14), 7291–7321. https://doi.org/10.5194/acp-14-7291-2014
- Painemal, D., & Zuidema, P. (2011). Assessment of MODIS cloud effective radius and optical thickness retrievals over the Southeast Pacific with VOCALS-REx in situ measurements. *Journal of Geophysical Research: Atmospheres*, 116. https://doi.org/10.1029/2011jd016155
- Szczodrak, M., Austin, P. H., & Krummel, P. B. (2001). Variability of optical depth and effective radius in marine stratocumulus clouds. *Journal of the Atmospheric Sciences*, 58(19), 2912–2926. https://doi.org/10.1175/1520-0469(2001)058<2912:voodae>2.0.co;2